
○ Users Plot

○ Accelerators Plot

• Crew Chief Summary:
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○  Further discussion about A49H.   Was not conditioned properly.
○ Shortened up the store and re-conditioned.
○ Store had problems going in.   The feeddown failure.
○ Initial was low and lifetime was sub=par
○ Beam intensity was also down.

• Notes from Run Coordinator:

□  LRF2 off - 10 minutes.
□ Klystron Debuncher - valve working find. 

� Summary:

□ Swtich over to other Kly deb. Pump.

� Requests:  

○ Linac

□  Downtime for bRF1 - all RF is up and running
□ Down in intensity due to losses, now back up. 

� Summary:

□ Work on transition. 

� Requests:  

○ Booster

□ HLRF - switch blower on RF 13 fixed.
□ Station 3 modulator fixed
□ Multibatch slipstacking on 2E 
□ Experts working on 2E

� Summary:

□  continue 2E studies

�  Requests:

○ Main Injector

Summary:

○ Pbar

• Machine Summaries:
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♦ Best Stacking = 16.49  mA/hr

♦ <Production> = 15.2 e-6/proton

♦ <Beam on Target> = 6.7 e12 

□ Pbar Stacking Numbers  

□ Bad emittance blowup - software problem - core momentum thermostat.   
Said the thermostat was on, but not.   Core got too thin

□ Target rotation stopped yesterday.  Corrosion problem in vault.   Gear box 
added to make more torque.  

□ Beam positions in P1 - P2 and AP1 between running NuMI and not running 
NuMI.

□

♦ Beam loss in beam lines

♦ HV202 - 8 GeV supply in the AP1 line.   Current readback was correct.  
Power cycled, then the beam went through.

5 Transfers

� Summary:

□ AP10 switch work. 

� Requests:

□ Conditioned A49H separator conditioned in wrong polarity.   Re-conditioned.   
Should help.

♦ Feeddown tripped on OC.

♦ F171K3 (proton inj kicker) - failed to fire, pre-fire. 

□ 4895 shot setup

� Summary:

□ Change out 465 for Feeddown. 
□ Kicker guys look at between stores.

� Requests:

○ Tevatron

□ Access MI31 with electricians - look at AC units to be installed. 
□ Shot to Tevatron, beam was not cooled.  Emittances were larger than usual. 
□ Cons test sequencer to unmine today.
□ Also work on other sequencers.

� Summary:

□ Cooling studies
□ Look at cogging and emittance growth. 

� Requests:

○ Recycler

□  No report

� Summary:

□

� Requests:

○ SY120

□ Dewatering pump #1 failed.    Being replaced by FESS
□ One of sump pumps not looking right.   Also being replaced. 

� Summary:

□

� Requests:

○ MiniBooNE

□ Access to fix water leak - 90min. Of cool down.   Ended up being 6.5 hours.   
RAW water leak fixed.

□

� Summary:

□
□

� Requests:

○ NuMI

□ 4893 - 92%

♦ Cooling water flow E side for Si.   Bubbles were seen there also.  

♦ Single channel calorimatry problems

♦ Forward muon system.

□ 2 hour controlled access.

□ 4895 - 92% 

� Summary:

Requests:

○ CDF
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□

� Requests:

□ Ran well
□ Access - cean cards for central muon system
□ New store - lost 15 minutes, then ok. 

� Summary:

□

� Requests:

○ D0

□

� Summary:

□

� Requests:

○ FESS

□ 2 failing roughers in TH. 

� Summary:

□

� Requests:

○ Mechanical

□

� Summary:

□

� Requests:

○ Other

� Terminated in middle evening.
� Kicker guys F17K3
� Feeddown card change. 

○ Summary:

�   

○ Requests:

• The Plan
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